\magnification=\magstep1
\nopagenumbers
\font\bigbf=cmbx10 scaled\magstep2
\centerline{\bigbf University of Utah}
\smallskip\centerline{\bigbf Math 1090, Section 4}
\smallskip\centerline{\bigbf Final Examination, December 16, 1999}
\bigskip\bigskip\bigskip\line{\bigbf Name:\ \hrulefill}
\bigskip\centerline{\vbox{\offinterlineskip\hrule 
\halign{$\vphantom{\bigg|}$\vrule#&\ \ \hfil#\hfil\ \ &\vrule#&\ \ 
\hfil#\hfil\ \ &\vrule#&\ \ \hfil#\hfil\ \ &\vrule#\cr 
&\bf Exercise&&\bf Total&&\bf Score&\cr
\noalign{\hrule}
\noalign{\vskip1pt}
\noalign{\hrule}
&1&&$10$&&&\cr
\noalign{\hrule}
&2&&$10$&&&\cr
\noalign{\hrule}
&3&&$10$&&&\cr
\noalign{\hrule}
&4&&$10$&&&\cr
\noalign{\hrule}
&5&&$10$&&&\cr
\noalign{\hrule}
&6&&$10$&&&\cr
\noalign{\hrule}
&7&&$10$&&&\cr
\noalign{\hrule}
&8&&$10$&&&\cr
\noalign{\hrule}
&9&&$10$&&&\cr
\noalign{\hrule}
&10&&$10$&&&\cr
\noalign{\hrule}
&11&&$10$&&&\cr
\noalign{\hrule}
&12&&$10$&&&\cr
\noalign{\hrule}
\noalign{\vskip1pt}
\noalign{\hrule}
&\bf Total&&120&&&\cr}
\hrule}}

\medskip\leftline{\bf Instructions}

\smallskip\item{1.}  Do not detach the pages of this exam. 

\smallskip\item{2.}  No books or notes are allowed. A list of formulae from
Chapter 6 is provided. You may use a calculator.

\smallskip\item{3.}  Use the back of the test sheets if scratch paper is needed. 

\smallskip\item{4.}  SHOW ALL YOUR WORK!

\bye

